Author's response to reviews

Title: Prevalence and Risk Factors for Non-use of Antenatal Care Visits: Analysis of the 2010 South Sudan Household Survey

Authors:

Ngatho S Mugo (n.mugo@hotmail.com)
Michael J Dibley (michael.dibley@sydney.edu.au)
Kingsley E Agho (k.agho@uws.edu.au)

Version: 6 Date: 2 February 2015

Author's response to Editor:

Title of MS: Prevalence and Risk Factors for Non-use of Antenatal Care Visits: Analysis of the 2010 South Sudan Household Survey

MS No.: 2012155437132335

Authors:

Ngatho S Mugo n.mugo@hotmail.com
Michael J Dibley michael.dibley@sydney.usyd.edu.au
Kingsley E Agho k.agho@uws.edu.au

Version: 5 Date: 02 February 2015

Author's response to Editor: In the manuscript the new information were highlighted in yellow. See below for the response.

Editorial Comments

Copyediting

- We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service.

Response: We agree and have copyedited the entire paper to improve the style of written English (see highlighted section in the attached manuscript).